Author's response to reviews

Title: Gated myocardial perfusion SPECT underestimates left ventricular volumes and shows high variability compared to cardiac magnetic resonance imaging - a comparison of four different commercial automated software packages

Authors:

Fredrik Hedeer (Fredrik.Hedeer@skane.se)
John Palmer (John.Palmer@skane.se)
Håkan Arheden (Hakan.Arheden@med.lu.se)
Martin Ugander (Martin.Ugander@med.lu.se)

Version: 8 Date: 18 May 2010

Author's response to reviews:

The revised manuscript has now undergone language editing by a native English speaker.

Sincerely

Martin Ugander